Author's response to reviews

Title: Modelling the cost-effectiveness of increasing alcohol taxes

Authors:

Matthijs van den Berg (matthijs.van.den.berg@rivm.nl)
Pieter HM van Baal (pieter.van.baal@rivm.nl)
Luqman Tariq (luqman.tariq@rivm.nl)
Jantine AJ Schuit (jantine.schuit@rivm.nl)
Ardine GA de Wit (ardine.de.wit@rivm.nl)
Rudolf T Hoogenveen (rudolf.hoogenveen@rivm.nl)

Version: 2 Date: 14 November 2008

Author's response to reviews:

MS: 6337856491955582

Research article

Modelling the cost-effectiveness of increasing alcohol taxes

Matthijs van den Berg, Pieter HM van Baal, Luqman Tariq, Jantine AJ Schuit, Ardine GA de Wit and Rudolf T Hoogenveen

Dear Editor,

We submitted our manuscript to BMC Public Health, but in her acceptance mail the editor asked us to publish the paper in BMC Medicine. We agreed with the request. In order to proceed, she asked to complete the submission process here.

Best regards,

Matthijs van den Berg